University of Waterloo STAT 332 – W. H. Cherry

#2.81a

Figure 2.10. UNSTRATIFIED POPULATIONS: Estimating a Proportion or a Frequency Section headings to provide a content overview

One-Stage EPSWOR of Individual Elements

1. Notation 2. Estimating^−P, the Respondent Population Proportion 3. Confidence Intervals for^−P, the Respondent Population Proportion

Example 2.10.1 Example 2.10.2 4. Confidence Intervals for −N^−P, the Respondent Population Number or Frequency

Example 2.10.3 Example 2.10.4 5. Appendix 1: Terminology Review 6. Appendix 2: Notation Review 7. References Appendix 8 of Figure 2.3 continued from page 2.50 Nine tables One diagram Nineteen equations

2023-08-20